Title: Heterotopic Ossification after Patellar Tendon Repair presenting in Trisomy 8 Mosaicism: a case report and literature review

Authors:

Austin W Chen (achen9@uic.edu)
Samuel J Chmell (schmell@uic.com)

Version: 4 Date: 5 July 2011

Author's response to reviews: see over
To the editors of the Journal of Medical Case Reports:

- Please restructure the Abstract into the following three sections: Introduction, Case presentation, and Conclusion. The abstract should be no longer than 350 words.

  The abstract has been restructured as advised.

- Please include the authors contribution section. We suggest the following kind of format (please use initials to refer to each author’s contribution): FC analyzed and interpreted the patient data regarding the hematological disease and the transplant. RH performed the histological examination of the kidney, and was a major contributor in writing the manuscript. All authors read and approved the final manuscript.

  The author’s contributions section has been added at the end.

- Please include the ethnicity of the patient in the case presentation and abstract section of the manuscript

  The ethnicity of the patient has been included in the appropriate areas.

Regards,
-Austin Chen

Austin Chen, MD
University of Illinois at Chicago
Department of Orthopaedic Surgery